

Beyond Dichotomy: The Curvilinear Impact of Employee Ownership on CEO entrenchment

Xavier Hollandts, Nicolas Aubert, Abdelmehdi Ben Abdelhamid, Victor Prieur

▶ To cite this version:

Xavier Hollandts, Nicolas Aubert, Abdelmehdi Ben Abdelhamid, Victor Prieur. Beyond Dichotomy: The Curvilinear Impact of Employee Ownership on CEO entrenchment. Management international = International management = Gestión internacional, 2017, 22 (2). halshs-01495427v1

HAL Id: halshs-01495427 https://shs.hal.science/halshs-01495427v1

Submitted on 7 Jun 2017 (v1), last revised 31 Dec 2018 (v2)

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Beyond Dichotomy:

The Curvilinear Impact of Employee Ownership on CEO entrenchment

Xavier Hollandts¹

Nicolas Aubert²³

Abdelmehdi Ben Abdelhamid⁴

Victor Prieur⁵

Abstract

Employee stock ownership gives employees a voice and therefore may have a major impact on corporate governance. Thus, employee stock ownership may be a powerful mean to protect CEOs from both market for corporate control and dismissal threat. In this paper, we examine the relationship between employee stock ownership and CEO entrenchment. Following the recent French legislative changes, we use a comprehensive panel dataset of the major French listed companies over the 2009-2012 period. We document inverted U shaped relationships between employee stock ownership and CEO entrenchment. Board employee ownership representation also plays a role and increases the inflexion points of these curvilinear relationship.

Keywords: employee stock ownership, corporate governance, CEO entrenchment.

1

¹ KEDGE BS & CRCGM, <u>xavierhollandts@kedgebs.com</u>. Corresponding author. Adress: 680 cours de la Libération, 33405 Talence Cedex, France.

² Aix Marseille Univ, CERGAM, Puyricard, France.

³ Inseec business school.

⁴ Auvergne University, CRCGM.

⁵ Paris-Dauphine University

Employee stock ownership (ESO from now) is a powerful tool sometimes presented as a way to promote shared capitalism (Kruse et al, 2010). When a firm makes some profits, part of them are usually plowed back and retained (self-financing), and some are shared among shareholders and, for a minor part, employees. With ESO schemes, employees are offered a part of profits, which allows them to increase their personal wealth, being better associated with the firm's success. The success of ESO mechanism is undeniable (Kim and Ouimet, 2014) and for instance, 23 millions of American employees (20% of total employee workforce - National Center for ESO, 2014) are employee owners. In France, we estimate to almost 3.5 millions of French employee owners⁶. This success can be analysed as an evidence of a "shareholder capitalism" since there are more and more shareholders in the world and employee owners take part of this worldwide and enduring phenomenon.

The decision of implementing and developing ESO often lies with management. Executive managers have a discretionary power to implement such schemes (Scholes and Wolfson, 1990). Managers have two major motivations to reward their employees with stock: to incentivize them to enhance corporate performance (Kim and Ouimet, 2014) and to retain them (Rauh, 2006). Indeed, the academic literature presents ESO as a two-edged sword. On one hand, ESO may be used as a reward management tool to enhance corporate performance through its

-

⁶ French workers have different ways to buy stocks of the company they work for. First, employees can invest in their company stocks simply by buying them directly from the financial market. The most common way to become an employee owner is to invest in a company savings plan (CSP or *Plan d'Epargne Entreprise* in French). Amounts that can be invested are the profit-sharing or gainsharing bonuses or the voluntary contributions. Investing in a PEE is financially rewarding for the employees for two reasons. First, money invested in the CSP by the employee can be matched by an employer contribution. Second, in the case of an employee stock purchase plan (ESPP), employees who buy stocks can benefit from a maximum discount of 20% on the stock price. The money invested in the CSP are frozen for a minimum of 5 years and they benefit from fiscal advantages. A major difference between the French CSP and the US ESOP is that the latter are mostly implemented by small and medium sized companies whereas a large majority of the French employee owners work for large listed companies. The ESOP also differs from the CSP in that it is only invested in company stocks. The CSP offers several alternative options. When setting up an ESPP, most French companies attribute the maximum discount combining them with matching contribution in company stock (see Rapp and Aubert, 2008 for more details and Ginglinger et al, 2011 for a focus on corporate governance).

incentive effects. Kaarsemaker (2006) summed up the literature on attitudinal effects of ESO and shows that employee owners are more satisfied, involved, productive, exhibit lower turnover and absenteeism rates and are likely to less free-ride On the other hand, it may be a management entrenchment mechanism via the potential collusion between employee owners and management. The impact of ESO on corporate governance is still being debated in the literature. One can argue that the presence of ESO has a positive impact because it decreases the overall level of asymmetric information of shareholders (Acharya et al, 2011). Besides, Faleye et al (2006) underline the "darks side" of ESO leading to management entrenchment and decreased shareholder value. Employee owners may protect executive managers with a friendly part of capital (Gamble, 2000). Thus, executive managers can reward employee owners with their protection, better job conditions or salaries (Cronqvist et al, 2009). In this configuration, employee owners can entrench themselves as CEOs do.

A large body of the literature examined the "dark side" of ESO (Gordon and Pound 1990; Chaplinsky and Niehaus, 1994; Park and Song, 1995; Gamble, 2000; Hellwig, 2000; Pagano and Volpin, 2005; Benartzi et al, 2007) but focuses mainly on the potential impact on corporate performance and governance. Our paper addresses also the question of the impact of ESO on governance by adding one new dimension of employee voice with their compulsory participation as board members in French boards. Some papers have examined the consequences of significant ESO on performance, risk, productivity or growth (Faleye et al, 2006; Park and Song 1995), on corporate takeovers (Pagano and Volpin 2005, Chaplinsky and Niehaus 1994), R&D intensity (Gamble, 2000). The test provided in such papers always examined the impact of employee share ownership on several dimensions of corporate governance and corporate performance. In our paper, we suggest to expand the consequences of ESO by considering the impact of "employee voice", namely the potential combination of employee share ownership and ESO board membership (Fauver and Fuerst, 2006). French law

and corporate governance system have evolved towards compulsory employee owners board membership at the end of 2000s and provide an opportunity to test such consequences of employee ownership's voice. Distinct from German codetermination regime (Fauver and Fuerst 2006), French corporate governance system is often presented as an hybrid system between Anglo-Saxon corporate governance regimes and Nordic and German corporate governance systems (Aste, 1999; Ginglinger et al, 2011). In this regard, "employee voice" may appear as a powerful mean for executive managers to increase their personal entrenchment (Faleye et al, 2006; Fauver and Fuerst, 2006; Gamble, 2000; Park and Song, 1995). Combining both the bright and dark side, we suggest that the impact of ESO and board employee ownership representation (BEOR from now) could be non-monotonic. In this paper, we investigate this issue by addressing the following research question: Does ESO affect CEO entrenchment? This impact can be split in two directions: Is ESO linked to CEO entrenchment? As ESO gives a voice to employee owners through potential board membership: Is board ESO participation linked to CEO entrenchment?

The novelty of our empirical results is to document a non-monotonic relationship between ESO and several measures of CEO entrenchment: CEO age, CEO tenure and CEO turnover. We also underline that BEOR increases the effect of ESO on CEO entrenchment by increasing the inflexion points of the curves. BEOR then plays a role in providing CEOs with an additional entrenchment tool. The remainder of the paper is organized as follows. Section 1 reviews the literature on the relationships between ESO, corporate governance and CEO entrenchment. Section 2 presents the methodology and the sample. Section 3 outlines results. Section 4 discusses findings, and section 5 concludes.

1. Literature review

ESO and firm performance

Broadly speaking, ESO is a management tool that can help to further enhance shared capitalism. Shared capitalism refers to "plans that tie worker pay or wealth to the performance of their own workplace" (Kruse et al, 2010; p. 5). In their book, Kruse et al (2010) investigate shared capitalism based on a representative sample of US workforce of more than 40,000 employees. They find evidence of the positive relationship between shared capitalism mechanisms including ESO and corporate performance. By giving employees a residual claim, it fosters positive job attitudes in the workplace, which therefore improves corporate performance. From an organizational perspective, ESO enhances cooperation and mutual monitoring within the workplace, increases productivity, decreases turnover and absenteeism. In his extensive literature review, Kaarsemaker (2006) concludes that most of the academic literature finds a positive relationship between ESO, employee attitudes and several firm performance measures. This is the bright side of ESO: a set of positive incentive mechanisms that foster various positive employee attitudes at work, thus improving corporate performance. In addition, ESO tends to retain human capital and to increase employee loyalty (Blair et al, 2000). Regarding corporate governance, ESO has unclear impacts due to the presence of bright and dark side effects.

ESO and internal governance

Indeed, as insider equity ownership, ESO has an intriguing role. ESO exhibits positive effects and can reduce CEOs' opportunistic behaviors since employee owners have an intimate knowledge of their organization (Acharya et al, 2011). By sharing information with external shareholders, they may mitigate asymmetric information. The internal governance of the firm can "force a self-interested CEO to act in a more public-spirited and far-sighted way" (Acharya et al, 2011; p. 689). In the same vein, Adams and Ferreira (2007) show that the natural proximity

with internal stakeholders as employees also contributes to a more efficient corporate governance since it helps to decrease the level of asymmetric information. For instance, Bova et al (2015) show that firms with ESO are more transparent and exhibit better disclosure. ESO also gives a strong incentive to monitor executive managers since a big part of employee owners' personal wealth directly depends on corporate decision makers (Blair, 1999; Kruse et al, 2010). Because ESO often represents a cost in terms of portfolio diversification (Yi Tsung et al, 2008), employee owners pay particular attention to the top management's decisions of their company. In this configuration, there is a potential alignment of interests for the main firm's stakeholders (employees, shareholders, executive managers) towards performance maximization. In fact, ESO helps to mitigate agency problems, mainly by reducing free-riding problems and incentivizes employees towards improved corporate performance (Kruse et al, 2010; Guedri and Hollandts, 2008). Nevertheless, a large body of the literature has shown that ESO also exhibits a dark side regarding corporate governance.

ESO and bilateral entrenchment: is there a dark side?

However, ESO also exhibits what Faleye et al (2006) call a dark side referring to its effects on corporate governance. Some authors argue that ESO is a powerful entrenchment tool because it reduces the probability of a takeover (Beatty, 1995)⁷. The "natural alliance" between employees and managers (Hellwig, 2000) encourages CEOs to establish implicit contracts with employee shareholders (Pagano and Volpin, 2005) in exchange for a friendly control of employee owners (Benartzi et al, 2007; Gamble, 2000). From this standpoint, ESO prevents an efficient market for corporate control (Chaplinsky and Niehaus, 1994; Park and Song, 1995).

_

⁷ In this paper, we consider that entrenchment has mainly negative effects on corporate governance (Morck et al, 1988; McConnell & Servaes, 1990) since the vast majority of corporate governance literature analyses it as a deviation from optimal configuration from an agency perspective (Fisman et al, 2013; Joseph et al, 2014 and Tihanyi et al, 2014). We acknowledge that there is a theoretical debate on the impact of CEO entrenchment since the seminal paper of Castanias and Helfat (1992) underlines that, in some circumstances, managerial entrenchment can create value for the firm and its shareholders.

Gordon and Pound (1990) consider that many ESOPs were established in the US in the late 1980s, explicitly to deter takeovers. Employee owners tend to vote against takeovers since they often go along with layoffs. Negative effects of ESO on corporate governance are well documented empirically. Indeed, ESO is more powerful than poison pills or golden parachutes (Chaplinsky and Niehaus, 1994) and other defensive mechanisms are less likely to be used when a firm already offers ESO schemes (Rauh, 2006; Park and Song, 1995). Besides, financial markets tend to react negatively to an ESOP's implementation announcement (Chang, 1990; Chang and Mayers, 1992), especially when they appear as a mean to entrench management (Cramton et al, 2008). Faleye et al (2006) argue that such a protection is more general, underlining that the dark side of ESO leads to a "bilateral entrenchment": CEOs may be more easily entrenched with ESO, but employees also tend to entrench themselves, benefitting from the CEOs' protection against layoffs (Atanassov and Kim, 2009). This is why ESO works as an insurance mechanism that ensures favorable employment policies and limits the risk of redundancies or wage cuts (Kim et al, 2014). In return, employee owners stand ready to help their CEO to fight off a potential takeover bid and his subsequent dismissal (Chaplinsky and Niehaus, 1994; Pugh et al, 1999). This mutual protection and commitment between executive managers and employee owners is set up on the terms of an implicit contract (Acharya et al, 2011; Atanassov and Kim, 2009; Pagano and Volpin, 2005). Employee owners can grant protection to their CEO since they control part of the capital and exercise their voting rights. As part of these implicit contracts, CEOs tend to implement policies aimed at maintaining or increasing levels of employment and wages (Faleye et al, 2006). In this regard, Cronqvist et al (2009) show that CEOs who wish to reach higher entrenchment levels, choose to better reward their employees in an attempt to reach "social peace". Therefore, shareholders pay this potential collusion with higher wages granted to employees, and a lower probability of receiving a takeover premium (Kim and Ouimet 2014, Faleye et al, 2006). Kim and Ouimet (2014, p. 1277) explicitly suggest that "management bribes employee with higher wages to garner worker support to thwart hostile takeover bids". This configuration, allows CEOs to have a "quiet life" by avoiding confrontations with external shareholders or the market for corporate control (Bertrand and Mullainathan, 2003). More precisely, Aubert et al (2014) consider that the incentive to increase managerial entrenchment by the mean of ESO may vary regarding to the corporate governance context. More specifically, it seems that the incentive for CEOs could vary depending on the corporate performance and on their level of managerial entrenchment. Increasing ESO would be less motivating for well-entrenched or/and high performer CEOs (Gregory-Smith et al, 2009).

ESO on CEO entrenchment: a curvilinear relationship?

The combination of two latent functions (namely the bright and dark side of ESO described above) could therefore leads to a non-monotonic relationship between ESO and CEO entrenchment. Haans et al (2016) deeply explained the rationale of quadratic relationships⁸. A U-shaped relationship may be conceptualized as two latent functions jointly making up a quadratic. While the latent functions are commonly not observable, they may be combined either additively or multiplicatively to explain a quadratic relationship that reveals the "net effect" of X on Y (Haans et al, 2016; p. 1178-79). In this paper, we consider that on one hand that the bright side of ESO mitigates agency problems, increases transparency and decreases the level of asymmetric information. But on another hand, ESO also exhibits a dark side. ESO can lead to a friendly control benefitting to the CEO. ESO can also be detrimental to external shareholders due to higher wages and less efficiency of corporate governance mechanisms. Thus, the net effect of the bright and dark side can be translated in a non-monotonic function between ESO and corporate governance efficiency. We suggest that this U-shaped relationship

_

⁸ We thank the anonymous reviewer that point out this very useful reference.

can be observed for one major corporate governance variable namely the CEO entrenchment. Two opposite forces can be at play and thus the relationship between ESO and CEO entrenchment can exhibit a U-shaped relationship. Thus, in this paper, we suggest that two opposite forces (the bright side – positive function and dark side - negative function) interact and thus a U-shaped relationship emerges due to the combination of two latent linear functions as described in Haans et al (2016). Regarding the shape of the relationship between ESO and governance, several recent papers underline that ESO exhibits non-linear relationships. Faleye et al (2006) test the impact of significant ESO (over 5% of share capital) on corporate performance, valuation, investment and risks⁹. They show that large ESO pushes away rather than towards shareholder-value maximization. Kim and Ouimet (2014) indicate that small ESOPs (under 5% of shares) tend to increase the size of the overall economic pie, benefitting both employees and shareholders. In their study, the positive effects of ESO occur at the threshold of 5% of the capital held by employees, and effects are much weaker for large ESOPs. Over this threshold, these gains are absorbed by higher wages (Cronqvist et al, 2009). Recent papers reveal the presence of a curvilinear relationship (Guedri and Hollandts, 2008) or at least of a 5% threshold (Faleye et al, 2006; Kim and Ouimet, 2014). Guedri and Hollandts (2008) also suggest that the presence of BEOR can moderates the relationship between ESO and performance and thus "employee voice" can play a crucial role. Also in the French context, Ginglinger et al (2011) test the impact of employee owners' representation on the board and find that it is at least value-neutral. In order to optimize corporate performance, the theoretical optimum level of ESO results from the combination of small levels of ESOPs that guarantee optimal incentives for employees and have low consequences in terms of free-riding problems (Kim and Ouimet, 2014; Aubert et al, 2014; Ginglinger et al, 2011; Guedri and Hollandts, 2008). Regarding the level of CEO entrenchment, Aubert et al, (2014) argue that ESO is a

⁹ In their seminal paper, Mc Connel & Servaes (1990) have shown a similar curvilinear relationship between corporate insiders (executive managers and board members) and performance (Tobin's Q).

powerful entrenchment mechanism for low-performing CEOs and may be less useful for highperforming or well-entrenched CEOs.

To sum up, the link between ESO and managerial entrenchment exhibits two opposing forces. As ESO exhibits bright side and dark side over corporate governance and performance, we suggest that two latent function are at play and can be translated in a curvilinear relationship between employee ownership and CEO entrenchment. Therefore, we state the following hypothesis:

H1: ESO exhibits a non-monotonic relationship with CEO entrenchment. ESO is positively related to CEO entrenchment across the low to moderate levels of ESO, and is negatively related to CEO entrenchment across the moderate to high levels of ESO.

Another consequence of ESO consists in giving a "voice" to employee owners since they can be offered seats on the board of directors. Hence, employee owners' representation could foster CEO entrenchment. In most European countries, employee owners can be represented on the board of directors or on the supervisory board (e.g. Germany and the co-determination system: Kim et al, 2014; Fauver and Fuerst, 2006). At the end of 2006, representation of employee owners on the board of directors became compulsory for French listed companies (Ginglinger et al, 2011) if employee owners collectively hold over 3% of shares, the general meeting of shareholders must give at least one seat to an employee owners' representative. BEOR is often viewed as an additional voice available to employee owners and as the recognition of ESO. When coupled with BEOR, ESO provides CEOs with an additional entrenchment tool (Gordon and Pound, 1990; Pugh et al, 1999). The natural alliance between employees and managers encourages CEOs to establish implicit contracts with employee owners (Pagano and Volpin, 2005), including eventual protection from employee owners during the meetings of the board

of directors (Guedri and Hollandts, 2008; Ginglinger et al, 2011). From the perspective of employee owners, it is much easier to lobby in favor of policy decisions that maximize their own interests if they have at least one seat on the board of directors. In exchange for support from employee owners (Atanassov and Kim, 2009; Fauver and Fuerst, 2006), CEOs would implement corporate policies aimed at increasing the level of employment and wage rate (Faleye et al, 2006; Cronqvist et al, 2009). As a consequence, employee owners and CEOs have a direct interest to set up mutual protection mechanisms. This is why there is a potential risk of "bilateral entrenchment" notably at board level (Ginglinger et al, 2011; Faleye et al, 2006). As BEOR is the consequence of significant levels of ESO (at least 3% of capital), potential implicit contracts can be established between employee owners board members and CEOs (Guedri and Hollandts, 2008). Therefore, we suggest that BEOR increases the potential relationship between ESO and CEO entrenchment. In other words, BEOR gives an additional mean of entrenchment to CEOs. Following Guedri and Hollandts (2008), we suggest that the presence of employee owners into the board is likely to facilitate friendly control and alliance between CEOs and employee owners' board members. Hence, BEOR seems to positively moderate the relationship between ESO and CEO entrenchment and thus the inflection point is likely to occur at lower level of ESO. We therefore put the following hypothesis:

H2: Board employee owners' representation positively moderates the non-monotonic relationship between ESO and CEO entrenchment.

2. Data and Methods

In this section, we present an empirical examination of the relationship between ESO, BEOR, and CEO entrenchment. We take into account variables that may affect this relationship and we test our research hypotheses based on the 120 largest French listed companies (SBF 120 index).

2.1.Data

Our panel comprises two types of data sources. First, we used the INSEAD OEE Data Services

(IODS) corporate governance, based on the 120 largest French capitalizations. This data was

previously used by Ginglinger et al (2011). Among other variables we use in this paper and that

we will detail further, the IODS dataset provides the proportion of equity hold by the employees

and the BEOR. We match corporate governance variables with data from Thomson Reuters

Eikon. Our dataset is longitudinal and covers the 2009-2012 period. It starts in 2009 due to the

gap between the 2006 act on BEOR and its actual enforcement. Indeed, the French government

established a two-year period to enforce the law (end of enforcement: 31/12/2008). Our data

include the CEO entrenchment variables between 2009 and 2012. The description of all the

variables included in the regressions is given in table 1.

Dependent variables: CEO entrenchment

The degree of managerial entrenchment was operationalized using three variables (Linck et al,

2008). The first one encompasses the number of years the CEO has filled his/her position within

the company (Hill and Phan, 1991). Several studies have shown that a high CEO longevity

makes him/her more resistant to internal and external pressures from different stakeholders.

This longevity allows the CEO to exert an influence, in his/her own favor, on the decision

process leading to the governance structure (Hermalin and Weisbach, 2000; Boone et al, 2007).

We also included the CEO's age variable. The CEO's age impacts his/her managerial

entrenchment strategy, especially when retirement becomes closer (Goyal eand Park, 2002).

We finally introduce a last variable measuring CEO entrenchment using turnover. This variable

takes the value of one if there is a CEO dismissal for a given year and 0 otherwise.

Independent variables: Employee stock and board employee ownership representation

12

ESO is a continuous variable that measures the level of ESO as the percentage of outstanding equity hold by employees for a given year (Aubert et al, 2014; Kruse et al, 2010). The percentage of ESO as a proxy for its size is crucial to examine the relationship between ESO and CEO entrenchment (Kim and Ouimet, 2014). BEOR is measured by the total number of employee owners divided by the size of the board i.e. the number of people seating on the board (Fauver and Fuerst, 2006). In French firms, board employee representation cannot exceed a third of board members, and this configuration cannot be assimilated to the co-determination system (Guedri and Hollandts, 2008). The French corporate governance system is sometimes seen as a "hybrid" pattern (Aste, 1999), midway between systems without any employee representation and the German pattern where two-tiered boards and employee representation are compulsory.

Control variables: governance, financial and sectoral characteristics

We introduce two kinds of control variables in our regression analyses: corporate governance variables and financial and sectoral characteristics. We first control for corporate governance characteristics: the ownership concentration, the French state ownership, the family ownership, the proportion of external directors on the board, the board size, the CEO duality and the presence of a two tiered board. These corporate governance variables may influence the focal relationship between the voice given to employees and CEO entrenchment. They are related to the ability of the CEO to be entrenched. The ownership concentration is measured by the largest shareholder (Hill and Snell, 1988). State and family ownership are two dummy variables taking the value of one if the French State of the founding family holds stocks of the company. These three ownership variables, by affecting management discretionary power, potentially affect CEO entrenchment. The board size is the total number of board members (Yermack, 1996) and is expected to increase entrenchment. The proportion of outside board members is also included

(Raheja, 2005) because it is assumed to decrease entrenchment. We follow Adams et al. (2010) who define outside board members as people with any current or past relationships with the focal firm (as employees, supplier, client or consultant)¹⁰. CEO duality is also measured. Duality is when the CEO and chairman positions are not hold by the same person. A last variable controls for the structure of the board. In France companies have the choice between a two tiered board or a unitary board structure. On another hand, we use variables controlling for financial and sectoral characteristics of the companies. The return on equity and the Tobin's Q account for accounting and financial performance respectively because they affect the managerial entrenchment strategy as predicted by the model of Aubert et al (2014). The sales variable captures size effects (Boone et al, 2007) and two dummies capture to which sector the company belongs. Sec1 is a dummy taking the value of 1 if the company belongs to the industrial sector (raw material, industry, consumption goods). Sec2 is a dummy taking the value of 1 if the company belongs to the service sector (health services, consumer services, telecommunication, services to communities).

[insert table 1 here]

2.2.Methods

We use panel GMM estimators to alleviate endogeneity concerns. Indeed, corporate governance research is subject to endogeneity issues since the investigated variables are endogenous by nature as it is extensively documented by Baghat and Jeffries (2005). GMM estimator was

¹⁰ In addition, this proxy of outsiderness encompasses the definition mainly used in French corporate governance codes (AFEP/MEDEF code, following Vienot and Bouton reports). Out of 120 listed companies, 108 made explicit reference to AFEP/MEDEF code to classify board members as outsiders.

developed in a series of papers (Arellano and Bond, 1991; Blundell and Bond, 1998) and improves fixed-effects OLS estimates when endogenous variables are included in regressions (Wintoki et al, 2012). Endogeneity can occur in OLS regression and may be the consequence of five different issues: error-in-variables (measurement error), autoregression, omitted variables, simultaneous causality and reverse causality (Wintoki et al, 2012; Semadeni et al, 2014). In each of these scenarios, OLS regression reports biased coefficients. Instead of estimating the true relationship between the independent variable and the dependent variable, OLS regression mistakenly includes the correlation between the independent variable and the error term in the estimation of the independent variables' coefficients. As most empirical corporate governance studies, our research model is likely to be affected by endogeneity. First, we can never totally exclude the risk of simultaneous causality. Second, we cannot also exclude that if ESO and BEOR affect CEO entrenchment, the reverse can also occur. In fact, CEO with long tenure and close to retirement can have developed ESO for years. Regarding unobservable heterogeneity (factors that affect both dependent and independent variables), OLS regression results can report spurious but statistically significant estimates (Wintoki et al, 2012). In the case of dynamic relation between an explanatory variable and past realizations of the dependent variable, a fixed-effect regression may be biased and the direction of the bias will be opposite of the dynamic relation. GMM panel estimator exploits the dynamic relationship inherent in our explanatory variables and the dynamic relation between our independent variables and CEO entrenchment. This problem is severe in corporate governance research and again GMM estimators can provide unbiased estimates (Wintoki et al, 2012).

[insert table 2 here]

[insert table 3 here]

3. Results

3.1.Descriptive statistics

Descriptive statistics are displayed in table 2. The median age of CEOs is 56 years and they hold their position for 4,71 years on average. The average percentage of equity hold by employees is 2,41%, which is higher than in previous studies: 1,62% for Guedri and Hollandts (2008) and 1,63% for Ginglinger et al (2011). This difference is due to the time window of our study which focuses on recent years whereas previous studies cover older periods, ending respectively in 2005 and 2008. The fact that ESO is, by essence, cumulative (Blair et al, 2000) means that firms have experienced continuous ESO schemes for the last years. The mean proportion of BEOR for the whole sample corresponds to 2.2% and only 34 firms (out of 110) experienced BEOR. Regarding control variables, the average ownership concentration is 29.05%, 32.77% and 13.84% have respectively family or state ownership. 41.34% of board members are external and the average board size is 12,63 members which is very close to figures reported by Ginglinger et al (2011). 72% of CEOs also hold the position of chairman of the board and 24% of companies exhibit a dual structure with a two-tiered board structure. Mean comparison tests are also displayed in the table 2. We use the median level of ESO (1.03% of the equity) as a reference point of our comparisons tests. We then compare companies with ESO above (>MED in the table 2) and below this level (<MED in the table 2). T-tests are significant for the following variables: ROE (+), Sales (-), ownership concentration (+), external directors (-), board size (-), board employee representation (-) meaning that companies with higher levels of ESO tend to have lower ROE, higher sales, lower ownership concentration, more external directors, higher board size and more employee owners on the board. The correlation matrix reported in table 3 does not show high correlations between exogenous variables with the exceptions of the correlation between board size and sales (0.49) and external directors and ownership concentration (-0.48). Although the first correlation seems obvious, the second suggests a substitution effect between ownership concentration that could be compensated by the presence of external directors.

3.2.Regression results

The data relates to 110 firms over a four-year period (2009–2012), we apply dynamic panel data econometrics techniques with robust standard errors (Greene, 2012). We report all the coefficients, standard errors and statistics in the tables 3 and 4.

[insert table 4 here]

Table 4 displays GMM regressions' estimates of ESO and control variables on CEO entrenchment measures: CEO age (models 1 to 3), CEO tenure (models 4 to 6) and CEO turnover (models 7 to 9). For each of these three dependent variables, we use a hierarchical approach where the first model only includes control variables, the second model introduces the ESO variable and the third model includes both the ESO and ESO squared variables in order to test the curvilinear relations. Overall results from models 3, 6 and 9 and the main novelty of this paper is to show the presence of curvilinear relationships between ESO and CEO entrenchment. We use the Lind and Mehlun's (2010) method to check the presence of curvilinear effects and to compute the inflexion points¹¹. We find that the inflexion points are the following: 11.73% for model 3, 17.77% for model 6, 11.94% for model 9. In the model 3, the relation between ESO and CEO age is U shaped whereas in models 6 and 9 measuring respectively the links between ESO and CEO tenure and between ESO and CEO turnover have

_

¹¹ Lind and Mehlun (2010) developed themselves a Stata module to test curvilinear relationships and compute the inflexion points of these relationships. For further details, see: http://econpapers.repec.org/software/bocbocode/s456874.htm

the shape of an inverted U. In the case of the regressions on CEO age, we cannot affirm that the curvilinear effect is validated because the coefficient associated to ESO is only significant at the 10% level. Regarding the regressions computing the direct effect of ESO on CEO entrenchment, both models 6 & 9 display significant coefficients with CEO tenure and turnover. Other coefficients associated to our control variables are significant. For the regressions on CEO age (models 1 to 3), the following coefficients are significant: external directors (+), board size (+), CEO duality (-), state (-), Tobin's Q (+) and sales (-). Regressions' coefficients associated to ownership concentration, dual structure, family ownership are not significant. The regressions on CEO tenure (models 4 to 6) have the following significant coefficients: ownership concentration (-), external directors (+), board size (+), CEO duality (+), dual structure (-), ROE (+), sales (-) and Tobin's Q (+). Regressions' coefficients associated to ownership concentration, family and state ownership are not significant. For the regressions on CEO turnover (models 7 to 9), the following coefficients are significant: ownership concentration (-), CEO duality (< for models 7 and 8 and >0 for model 9), family ownership (-), ROE (-) and Tobin's Q (+).

[insert table 5 here]

Table 5 displays GMM regressions' estimates of BEOR and control variables on CEO entrenchment measures: CEO age (model 10), CEO tenure (model 11) and CEO turnover (model 12). In these regressions, we test the moderating effect of BEOR on the previously investigated relationships between ESO and CEO entrenchment. We do not document a direct significant relationship between BEOR and the three variables of entrenchment i.e. none of the coefficients associated to BEOR is significant. We then test other specifications including interaction effects between ESO and BEOR following Guedri and Hollandts (2008). These

authors suggest that BEOR moderates the curvilinear relationship between ESO and performance. They do not validate this hypothesis. Although our approach is very similar from a technical point of view, it differs from theirs in two ways: we do not use performance as the dependent variable but we focus directly of entrenchment and we use BEOR instead of board employee representation of employees whether they are employee owners or not. In this setting, we find significant relationships between our interaction terms (BEOR×ESO and BEOR×ESO2) and CEO tenure on one hand and CEO turnover on another. Coefficients associated to the BEOR variable are not significant when CEO age is the dependent variable. Again, the results suggest curvilinear relations. The Lind and Mehlun's (2010) method confirms curvilinear effects and compute the following inflexion points: 11.79% for model 10 (11.73%) without including BEOR), 17.83% for model 11 (17.77% without including BEOR), 12.49% for model 12 (11.94% without including BEOR). The results show that BEOR positively moderates the effect of ESO on CEO tenure and CEO turnover. The interpretation of these results is that BEOR decreases the inflexion points of the curvilinear relationship for CEO age (but the coefficients associated to BEOR in this model are not significant) and increases the inflexion points for CEO tenure and turnover (with significant coefficients). Other coefficients associated to our control variables are significant. For the regressions on CEO age (model 10), the following coefficients are significant: ownership concentration (-), dual structure (-), family (+), ROE (+) and Tobin's Q (-). Regressions' coefficients associated to the other variables are not significant. The regressions on CEO tenure (models 11) have the following significant coefficients: external directors (+), board size (+), dual structure (-), family (+), ROE (+) and Tobin's Q (-). Regressions' coefficients associated to the other variables are not significant. For the regressions on CEO turnover (model 12), the following coefficients are significant: ownership concentration (-), dual structure (-), family (-), ROE (-) and Tobin's Q (-).

3.2 Robustness checks

We run additional tests to check the sensitivity of our analyses to alternative explanations. Endogeneity is an important concern for research in corporate governance and surely affect the relationship between ESO, BEOR and CEO entrenchment. But it is treated thanks to the GMM regressions. Also, some variables potentially affect the relation between ESO and BEOR and CEO entrenchment. We include three additional variables and run the GMM regressions: total debt, CEO social capital and CEO human capital. Literature in corporate finance argues that debt may play a disciplinary role by reducing management discretionary power (Jensen, 1986). Previous papers in corporate governance suggest that CEO human and social capital affect corporate governance (Johnson et al, 2012). We measure the total debt as the total debt issued by the firm in Euros (source: Thomson Reuters Eikon), the human capital with a dummy variable taking the value of 1 when a CEO graduated from Ecole Nationale d'Administration, Ecole Polytechnique or Hautes Etudes Commerciales and 0 otherwise (source: IODS), the social capital is the total number of board of directors mandates in French listed companies hold by the CEO. The coefficients associated with these three variables are not significant in most of the models. We therefore choose not to include these variables in the reported tables. We interpret these results as a consequence of the variance of CEO entrenchment explained by these variables as already captured by other variables. Debt is very correlated with other financial and accounting characteristics such as the size of the company, the ROE and the Tobin's Q already in our regressions. CEO human and social capital are very related to other variables we already have in our regressions such as ownership variables or governance variables. We also tried alternative combinations of variables and we only keep and report the models that provides the best results in terms of model fit and significance.

4. Discussion

In this paper, we focus on the relationship between employee stock ownership (ESO), its potential representation into the board (BEOR) and corporate governance. Only few papers have suggested that ESO can be part of a broader form of employee power namely employees' voice. In the French context, employee owners have a fraction of capital but can also be (compulsorily) appointed at board level. Guedri and Hollandts (2008) and Ginglinger et al (2011) have examined the impact of such representation and found mixed results. Thus, there is a theoretical and empirical debate in the corporate governance literature about the impact of ESO. We take into account the two literature streams and suggest that ESO and BEOR exhibit two opposite forces. The "bright" side suggests that ESO and its potential board membership increases transparency, decreases the level of asymmetric information and ensures more efficient corporate governance mechanisms. ESO is often viewed as a management and financial tool that increases corporate performance, decreases labor conflicts and finances corporate growth. But as suggested by the other stream, ESO can be seen as a powerful entrenchment tool that helps CEOs to extend their tenure and repel threats of dismissal (through market for corporate control). In addition, significant ESO levels push away rather towards shareholder-value maximization (Faleye et al, 2006). Combining two latent and countervailing linear functions (see Haans et al, 2016 for further details), we suggest that the bright and dark side of ESO translate into a non-monotonic relationship with our focal variable, namely CEO entrenchment. In this paper, we focus mainly on CEO entrenchment since the corporate governance literature has shown that the CEO function is central for corporate governance efficiency (Joseph et al, 2014; Tihanyi et al, 2014).

To our knowledge, our paper is the first to empirically test the focal relationship in the French context since the adoption of (legal) compulsory BEOR. France offers a unique perspective to test a mix of voluntary ESO mechanisms (French ESO schemes are based on voluntary

participation) and compulsory mechanism (BEOR). The French context represents a hybrid, and somewhat unique case, apart from US corporate governance system (with no board employee board membership) or German codetermination regime (with compulsory tradeunion board membership (Aste, 1999; Ginglinger et al, 2011; see also Tihanyi et al, 2014 for a discussion). Nevertheless, we believe that our results are generalizable to other advanced economies for several reasons. Ginglinger et al (2011) underline that French listed companies, are comparable firms with in all European and North American countries since these firms are highly visible companies operating in competitive global industries. In addition, the capital breakdown of French listed companies is diversified and internationalized as non-French investors own over half of the stock of private companies (Ginglinger et al, 2011). The unique mix of ESO participation and compulsory board employee owners' appointments in France give us the opportunity to test two dimensions of employees' power (by collectively holding a fraction of capital and/or a seat into the board of directors).

Our paper also contributes to the empirical literature since significant evidence of the relationship between employee ownership's voice and CEO entrenchment is highlighted. Our models reveal non-monotonic relationships between ESO and CEO entrenchment (measured by CEO age, CEO tenure and CEO turnover). Additional tests also show that BEOR moderates the ESO – CEO entrenchment relationship, for two dimensions of CEO entrenchment (CEO tenure and turnover), by fostering the impact of ESO on CEO entrenchment. Results from model 3, 6 and 9 provide a strong support for our first hypothesis. We mainly observed a significant inverted U-shaped relationship between ESO and CEO entrenchment. Results from models 6 and 9 underline that ESO could be an interesting tool for CEOs who want to enhance their entrenchment. Our results also show a curvilinear (U-shaped) relationship between ESO and CEO age. Taken together, our results suggest that ESO can be an effective entrenchment tool, up to a certain point. Beyond the inflexion points, CEOs must be well entrenched and/or

other governance mechanisms may be at play. Regarding CEO age, our results suggest that ESO could be less interesting for CEOs near retirement but could be more profitable for CEOs that go beyond the retirement age. Nevertheless, further research is clearly needed to go deeper in the understanding of such an apparent "paradox".

Results from models 10 to 12 provide also a strong support for our second hypothesis. For these models, our results highlight the presence of significant and positive moderating impact of BEOR. Our results show that ESO and BEOR contribute to explain our variables capturing CEOs' entrenchment. Our results underline the combined effect of employee voice consisting in stock ownership and board membership. By giving additional voice (board membership), ESO offers an additional tool that give a potential entrenchment tool to CEOs. Theoretically, employee owners may more easily enter into implicit contracts with CEOs (Pagano and Volpin, 2005) in return for favorable corporate policies (Cronqvist et al, 2009; Acharya et al, 2011). Broadly speaking, recent papers in the literature point out non-monotonic relationships between ESO and corporate performance or corporate governance variables (Faleye and al, 2006; Guedri and Hollandts, 2008; Kim and Ouimet, 2014) and our paper gives new evidence on the mixed impact of ESO. Our purpose was to fill a gap in the literature, showing that ESO affects CEO entrenchment, and not only in specific cases of takeover threats. ESO is mainly considered in the literature as a "friendly" fraction of capital for CEOs (Gamble, 2000) but few papers have tried to examine empirically the impact on corporate governance except in special cases such as takeover and broadly the market for corporate control (Kim and Ouimet, 2014). Our paper tries to test the impact of ESO in regular corporate governance context. Furthermore, we provide additional test of BEOR and highlight the moderating impact of such board representation.

Overall results (from direct and moderating effects) are interpreted as an evidence of bilateral entrenchment since employee owners are incentivized to maintain implicit contracts with their

CEO until he/she retires. To the best of our knowledge, our results could also be interpreted with regard to former empirical studies in the French context. Guedri and Hollandts (2008) have shown the curvilinear impact of ESO on firm performance for a larger index of listed companies (SBF 250). However, they do not find support for BEOR whereas our results suggest that BEOR plays a role. Moreover, our results somewhat contrast with findings of Ginglinger et al (2011). Indeed, their observations show that board employee representation could at least appear value-neutral, and in some circumstances value-enhancing for firm performance. They found "that directors elected by employee shareholders increase firm valuation and profitability, but do not significantly impact corporate payout policy" (Ginglinger and al, 2011; p. 868).

Our results highlight the mixed impact of ESO. Some authors have stressed the dark side of employee voice regarding corporate policies (Chen et al, 2012; Faleye et al, 2006) whereas employee voice may also be seen as an internal governance system (Acharya et al, 2011) that contributes to increase the level of transparency (Bova et al, 2015). From a corporate governance perspective, recent papers argue that ESO and employee voice still play a major role in terms of managerial entrenchment (Aubert et al, 2014). We do not directly test the impact on corporate performance or profitability but our results show that for corporate governance considerations, ESO exhibits mixed impacts since it contributes to foster CEO entrenchment in some circumstances. Thus our paper contributes to our overall understanding of the full consequences of ESO.

Combining overall results in the literature, we assert, as Fauver and Fuerst (2006), that ESO may enhance productivity, corporate performance and firm value (Kaarsemaker, 2006), though, in some cases, excessive levels of ESO may have a dark side, with potential negative impacts on CEO entrenchment. This notion of excessive levels of ESO is reflected in recent literature, suggesting that an optimal balance should be reached (Aubert et al, 2014; Kim and Ouimet, 2014; Guedri and Hollandts, 2008; Faleye et al, 2006). Our results, showing non-mononotic

relationships, give new evidence on the likelihood of optimal configuration of ESO, regarding corporate performance and corporate governance considerations.

Our results have also managerial implications. In France, as in many developed countries, ESO is increasing and is well developed. For example, 23 Millions of American are employee owners (20% of total employee workforce - National Center for ESO, 2014). In France, 3.5 Millions of French employees hold their company stocks. Thus, we need to clearly examine the full consequences of ESO on corporate governance since this phenomenon is still increasing. If the balance shifts in the right (bright) side, main stakeholders will give strong support to such ESO schemes. If not, we can expect that executive managers and main shareholders will prefer to slow down its current development. In our sample, the mean for firms with ESO is 2.4% of equity, not far from the 3% threshold of compulsory BEOR. Thus, external shareholders must carefully examine full consequences of ESO development within the firm and its capital breakdown.

Even though this paper contributes to the literature on ESO and corporate governance, it has several limitations. First, our study is limited to the largest French listed companies by focusing on the 120 biggest capitalizations. We acknowledge that ESO can be implemented in non-listed firms but for France, 99,6% of ESO is concentrated in listed companies (DARES – Statistics Institute of the French Ministry of Labour). Another interesting point for the SBF 120 is that this index is well balanced since almost half of this index is composed with companies that have implemented ESO schemes. Second, our study only spans on four years since we precisely start our study the year BEOR became compulsory. Consequently, our results must be interpreted cautiously. We think that the continuous development of ESO will help to confirm our results.

Further research is nevertheless needed to facilitate deeper understanding of what would be the potential combination of overall employee voice (through ESO) and BEOR. Except for

Germany and the co-determination system, which is well documented in the literature (Fauver and Fuerst, 2006; Kim et al, 2014), we obtained fragmented empirical results based on hybrid cases such as the French system (Guedri and Hollandts, 2008; Ginglinger et al, 2011). Some authors explore the potential combination of employee voice in terms of shareholding power and board representation. Another interesting stream in corporate governance literature can focuses on the employee power into the board. For instance, a recent paper from Balsmeier et al (2013), using game-theoretical and political models give us an opportunity to examine power relationships in the boardroom in the presence of employee (ownership) representation. To sum up, it seems obvious that in the current context a substantial part of employee voice lies in shareholding power, since board employee representation is modest¹². Nevertheless, comparisons with the current German co-determination system should be made with caution because the French institutional and legal environment are different, and results obtained with German parity firms should be interpreted taking into account their specific context.

An important avenue for future research obviously regards the financial literature. CEOs are strongly incentivized to establish implicit contracts with employee owners (Hellwig, 2000; Pagano and Volpin, 2005). Recent papers (Aubert et al, 2014) underline the fact that ESOs may be a powerful tool for CEOs in addition to being a financing mean, which can be attractive both for low-performing CEOs or newly appointed ones. Further research is also required regarding the determinants of corporate financing policies (CEO decision-making). Is ESO an optimal tool allowing to finance corporate growth and helping guard CEOs and firms against hostile takeovers?

 $^{^{12}}$ Only 34 firms on 110 have at least one board employee member; mean of board employee representation = 2,14 with board size mean at 14,63.

5. Conclusion

This study examines how ESO and BEOR affects CEO entrenchment. The French case is interesting in that it combines a mandatory board employee representation with a widespread ESO culture among listed firms. A comprehensive sample was used to assess the impact of ESO and BEOR on CEO entrenchment. Findings indicate that ESO exhibits non-monotonic (curvilinear) relationships with CEO entrenchment. Other tests underline that BEOR plays a substantial role by providing CEOs an additional entrenchment tool. Overall results show that ESO may be seen as a double-edged sword. On one hand, previous research has clearly documented a positive impact at the micro-level of organizations (implication, satisfaction, productivity, psychological ownership, etc.). On the other hand, corporate governance literature questions the impact of employee voice on corporate governance and corporate valuation. Acharya et al (2011) argue that a model of "internal governance", where subordinate managers monitor top management, may mitigate agency problems. However, ESO and board-level employee representation may operate as an insurance mechanism (Kim et al, 2014) that ensures the enforcement of implicit contracts (Pagano and Volpin, 2005).

Following Fauver and Fuerst (2006), we argue that excessive levels of employee voice, tend to impact labor itself, which becomes an agency cost. Indeed, employees pursue their own interests, exert their influence to maximize payroll rather than stock price, creating a situation in which monitors themselves need to be monitored. This concept of "excessive" employee voice has been described in the late literature, showing that the development of ESO must be carefully managed.

References

- Acharya, V. V., Myers, S. C., & Rajan, R. G. (2011). The Internal Governance of Firms. *Journal of Finance*, 66(3), 689-720. doi:10.1111/j.1540-6261.2011.01649.x
- Adams, R. B., & Ferreira, D. (2007). A theory of friendly boards. The Journal of Finance, 62(1), 217-250.
- Adams, R. B., Hermalin, B. E., & Weisbach, M. S. (2010). The role of boards of directors in corporate governance: A conceptual framework and survey. *Journal of Economic Literature*, 48(1), 58-107.
- Arellano, M., & Bond, S. (1991). Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations. *The review of economic studies*, 58(2), 277-297.
- Aste, L. J., 32, 1. (1999). Reforming French corporate governance: A return to the two-tier board. *George Washington journal of international law ans economics*, 32(1).
- Atanassov, J., & Kim, E. H. (2009). Labor and Corporate Governance: International Evidence from Restructuring Decisions. *Journal of Finance*, 64(1), 341-374. doi:10.1111/j.1540-6261.2008.01436.x
- Aubert, N., Garnotel, G., Lapied, A., & Rousseau, P. (2014). Employee ownership: A theoretical and empirical investigation of management entrenchment vs. reward management. *Economic Modelling*, 40, 423-434. doi:10.1016/j.econmod.2013.12.011
- Balsmeier, B., Bermig, A., & Dilger, A. (2013). Corporate governance and employee power in the boardroom: An applied game theoretic analysis. *Journal of Economic Behavior & Organization*, 91, 51-74.
- Beatty, A. (1995). The cash flow and informational effects of employee stock ownership plans. *Journal of Financial Economics*, 38(2), 211-240.
- Benartzi, S., Thaler, R. H., Utkus, S. P., & Sunstein, C. R. (2007). The law and economics of company stock in 401 (k) plans. *Journal of law and economics*, 50(1), 45-79.
- Bertrand, M., & Mullainathan, S. (2003). Enjoying the Quiet Life? Corporate Governance and Managerial Preferences. *Journal of Political Economy*, 111(5), 1043-1075.
- Bhagat, S., & Jefferis, R. H. (2005). The econometrics of corporate governance studies: Mit Press.
- Blair, M. M., Kruse, D. L., & Blasi, J. R. (2000). Employee ownership: an unstable form or a stabilizing force? In M. M. K. Blair, Thomas A. (Ed.), *The new relationship: human capital in the American corporation* (pp. 241-298). Washington D.C.: Brookings Institution Press.
- Blair, M. M., & Roe, M. J. (Eds.). (1999). *Employees and corporate governance*. Washington, DC: Brookings Institution Press.
- Blundell, R., & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. *Journal of econometrics*, 87(1), 115-143.
- Boone, A. L., Casares Field, L., Karpoff, J. M., & Raheja, C. G. (2007). The determinants of corporate board size and composition: An empirical analysis. *Journal of Financial Economics*, 85(1), 66-101. doi:10.1016/j.jfineco.2006.05.004
- Bova, F., Dou, Y., & Hope, O. K. (2015). Employee ownership and firm disclosure. *Contemporary Accounting Research*, 32(2), 639-673.
- Castanias, R. P., & Helfat, C. E. (1992). Managerial and windfall rents in the market for corporate control. *Journal of Economic Behavior & Organization*, 18(2), 153-184.
- Castanias, R. P., & Helfat, C. E. (2001). The managerial rents model: theory and empirical analysis. *Journal of Management*, 27, 661-678.
- Chang, S. (1990). Employee stock ownership plans and shareholder wealth: an empirical investigation. *Financial Management*, 19(1), 48-58.
- Chang, S., & Mayers, D. (1992). Managerial vote ownership and shareholder wealth: evidence from employee stock ownership plans. *Journal of Financial Economics*, 32(1), 103-131.
- Chaplinsky, S., & Niehaus, G. (1994). The role of ESOPs in takeover contests. *Journal of Finance*, 49(4), 1451-1470.
- Chen, H., Kacperczyk, M., & Ortiz-Molina, H. (2012). Do Nonfinancial Stakeholders Affect the Pricing of Risky Debt? Evidence from Unionized Workers. *Review of Finance*, 16(2), 347-383. doi:10.1093/rof/rfq028
- Cramton, P., Mehran, H., & Tracy, J. (2008). ESOP Fables: The Impact of Employee Stock Ownership Plans on Labor Disputes.
- Cronqvist, H., Heyman, F., Nilsson, M., Svaleryd, H., & Vlachos, J. (2009). Do Entrenched Managers Pay Their Workers More? *Journal of Finance*, 64(1), 309-339. doi:10.1111/j.1540-6261.2008.01435.x

- Faleye, O., Mehrotra, V., & Morck, R. (2006). When labor has a voice in corporate governance. *Journal of Financial & Quantitative Analysis*, 41(3), 489-510.
- Fauver, L., & Fuerst, M. E. (2006). Does good corporate governance include employee representation? Evidence from German corporate boards. *Journal of Financial Economics*, 82(3), 673-710. doi:10.1016/j.jfineco.2005.10.005
- Fisman, R. J., Khurana, R., Rhodes-Kropf, M., & Yim, S. (2013). Governance and CEO turnover: Do something or do the right thing? *Management Science*, 60(2), 319-337.
- Gamble, J. E. (2000). Management commitment to innovation and ESOP stock concentration. *Journal of Business Venturing*, 15(5-6), 433-447.
- Ginglinger, E., Megginson, W., & Waxin, T. (2011). Employee ownership, board representation, and corporate financial policies. *Journal of Corporate Finance*, 17(4), 868-887. doi:10.1016/j.jcorpfin.2011.03.005
- Gordon, L. A., & Pound, J. (1990). ESOPs and corporate control. *Journal of Financial Economics*, 27(2), 521-551.
- Goyal, V. K., & Park, C. W. (2002). Board leadership structure and CEO turnover. *Journal of Corporate Finance*, 8(1), 49-66.
- Greene, W. H. (2012). Econometric analysis (7th ed., International ed. ed.). Boston: Pearson.
- Gregory-Smith, I., Thompson, S., & Wright, P. W. (2009). Fired or retired? A competing risks analysis of chief executive turnover. *the economic Journal*, *119*(536), 463-481.
- Guedri, Z., & Hollandts, X. (2008). Beyond dichotomy: the curvilinear impact of employee ownership on firm performance. *Corporate Governance: An International Review*, 16(5), 460-474. doi:10.1111/j.1467-8683.2008.00703.x
- Haans, R. F., Pieters, C., & He, Z. L. (2015). Thinking about U: theorizing and testing U-and inverted U-shaped relationships in strategy research. *Strategic Management Journal*.
- Hellwig, M. (2000). On the economics and politics of corporate finance and corporate control. In X. Vives (Ed.), *Corporate Governance : Theoretical and Empirical Perspectives*. Cambridge, UK: Cambridge University Press.
- Hermalin, B. E., & Weisbach, M. S. (2000). *Boards of Directors as an Endogenously Determined Institution:* A Survey of Economic Literature. Working Paper.
- Hill, C. W. L., & Phan, P. (1991). CEO tenure as a determinant of CEO pay. *Academy of Management Journal*, 34(3), 707-717. doi:10.2307/256413
- Hill, C. W. L., & Snell, S. A. (1988). External control, corporate strategy, and firm performance in research-intensive industries. *Strategic Management Journal*, *9*(6), 577-590.
- Jensen, M. C. (1986). Agency cost of free cash flow, corporate finance, and takeovers. *Corporate Finance, and Takeovers. American Economic Review*, 76(2).
- Johnson, S. G., Schnatterly, K., & Hill, A. D. (2012). Board composition beyond independence social capital, human capital, and demographics. *Journal of Management*, 0149206312463938.
- Joseph, J., Ocasio, W., & McDonnell, M.-H. (2014). The structural elaboration of board independence: Executive power, institutional logics, and the adoption of CEO-only board structures in US corporate governance. *Academy of Management Journal*, 57(6), 1834-1858.
- Kaarsemaker, E. (2006). Employee ownership and its consequences: synthesis-generated evidence for the effects of employee ownership and gaps in the research literature. University of York.
- Kim, E., Maug, E., & Schneider, C. (2011). Labor Representation in Governance as an Insurance Mechanism: European Corporate Governance Institute.
- Kim, E. H., Maug, E. G., & Schneider, C. (2014). Labor representation in governance as an insurance mechanism. *European Corporate Governance Institute (ECGI)-Finance Working Paper*(411).
- Kim, E. H., & Ouimet, P. (2014). Broad-based employee stock ownership: motives and outcomes. *Journal of Finance*, 69(3), 1273-1319. doi:10.1111/jofi.12150
- Kruse, D., Freeman, R. B., & Blasi, J. R. (2010). Shared capitalism at work: employee ownership, profit and gain sharing, and broad-based stock options. *National Bureau of Economic Research conference report*. Chicago: The University of Chicago Press,.
- Linck, J. S., Netter, J. M., & Yang, T. (2008). The determinants of board structure. *Journal of Financial Economics*, 87(2), 308-328. doi:10.1016/j.jfineco.2007.03.004

- Lind, J. T., & Mehlum, H. (2010). With or without U? the Appropriate Test for a U-Shaped Relationship. Oxford bulletin of economics and statistics, 72(1), 109-118.
- McConnell, J. J., & Servaes, H. (1990). Additional evidence on equity ownership and corporate value. *Journal of Financial Economics*, 27(2), 595-612.
- Pagano, M., & Volpin, P. F. (2005). Managers, workers, and corporate control. *Journal of Finance*, 60(2), 841-868. doi:10.1111/j.1540-6261.2005.00748.x
- Park, S., & Song, M. H. (1995). Employee stock ownership plans, firm performance, and monitoring by outside blockholders. *Financial Management*, 24(4), 52-65.
- Pugh, W. N., Oswald, S. L., & Jahera Jr., J. S. (1999). ESOPs, Takeover Protection, and Corporate Decision-Making. 23(2), 170-185.
- Raheja, C. G. (2005). Determinants of Board Size and Composition: A Theory of Corporate Boards. *Journal of Financial & Quantitative Analysis*, 40(2), 283-306.
- Rapp, T., & Aubert, N. (2011). Bank employee incentives and stock purchase plans participation. *Journal of Financial Services Research*, 40(3), 185-203. doi:10.1007/s10693-011-0104-0
- Rauh, J. D. (2006). Own company stock in defined contribution pension plans: A takeover defense? *Journal of Financial Economics*, 81(2), 379-410. doi:10.1016/j.jfineco.2005.07.004
- Scholes, M. S., & Wolfson, M. A. (1990). Employee stock ownership plans and corporate restructuring: myths and realities. *Financial Management*, 19(1), 12-28.
- Semadeni, M., Withers, M. C., & Trevis Certo, S. (2014). The perils of endogeneity and instrumental variables in strategy research: Understanding through simulations. *Strategic Management Journal*, 35(7), 1070-1079.
- Tihanyi, L., Graffin, S., & George, G. (2014). Rethinking governance in management research. *Academy of Management Journal*, 57(6), 1535-1543.
- Wintoki, M. B., Linck, J. S., & Netter, J. M. (2012). Endogeneity and the dynamics of internal corporate governance. *Journal of Financial Economics*, 105(3), 581-606.
- Yermack, D. (1996). Higher market valuation of companies with a small board of directors. *Journal of Financial Economics*, 40(2), 185-211.
- Yi Tsung, L., Yu-Jane, L., & Ning, Z. (2008). The Costs of Owning Employer Stocks: Lessons from Taiwan. *Journal of Financial & Quantitative Analysis*, 43(3), 717-740.

Appendix

Table 1 : Description of variables

Variable name	riable name Source Description			
Dependent varia	bles: CEO entrenchn	nent		
Dependent varia	IODS corporate	CEO age is the age of CEO	Years	
CEO age	governance	ello ago is the age of ello	Tours	
020 ugu	IODS corporate	CEO tenure is the time the CEO has spent in position	Years	
CEO tenure	governance	elections are time the election as spent in position	10015	
CEO turnover	IODS corporate	CEO turnover indicates if a CEO dismissal happens (1) or	Binary 0/1	
	governance	not (0)	•	
Independent var	riables: ESO and ESO) representation on the boards		
ESO	IODS corporate	ESO is the percentage of shares held by employees.	Percentage	
	governance	1 0 , 1 ,	C	
BEOR	IODS corporate	BEOR is the proportion of employee owners seating on the	Percentage	
	governance	board	C	
Control variable	s: corporate governa	nce and financial characteristics		
Own.	IODS corporate	Own. Concentration measures the percentage of equity held	Percentage	
concentration	governance	by the largest shareholder	rereasinge	
External	IODS corporate	External director is the number of external directors divided	Percentage	
directors	governance	by the total board members	rereasinge	
Board size	IODS corporate	Board size is the total number of board members	Number	
Board Size	governance	Board Size is the total lighteer of board memoris	Trainion	
CEO duality	IODS corporate	CEO duality is the dummy variable taking the value of 1 if	Binary 0/1	
CEO ddanty	governance	the CEO is also the chairman and 0 otherwise	Dinary of 1	
Dual Structure	IODS corporate	Dual structure takes the value of 1 if the company has a two	Binary 0/1	
Daar Structure	governance	tiered board structure and 0 otherwise	Billary of I	
State	IODS corporate	State is the dummy variable taking the value of 1 if the state	Binary 0/1	
State	governance	holds shares of the company and 0 otherwise	Diliary 0/1	
Family	IODS corporate	Family is a dummy variable taking the value of 1 if the	Binary 0/1	
1 anniy	governance	company is a duminy variable taking the value of 1 if the	Dillary 0/1	
ROE	Thomson Reuters	ROE is the return on equity	Log of a percentage	
ROL	Eikon	ROL is the return on equity	Log of a percentage	
Sales	Thomson Reuters	Sales is the total sales of the company in euros	Log of the amount in	
Saics	Eikon	Sales is the total sales of the company in curos	Euros	
Tobin's Q	Thomson Reuters	Tobins' Q is defined as the market value of equity at the end	Proportion	
Toolii s Q	Eikon	of the fiscal year plus the book value of assets minus the	Troportion	
	EIKOII	book value of equity, all divided by the book value of assets		
Sec1	Thomson Reuters	Sec1 is a dummy taking the value of 1 if the company	Binary 0/1	
5001	Eikon	belongs to the industrial sector (raw material, industry,	Dilially 0/1	
	EIKOII	consumption goods)		
Saa2	Thomson Reuters		Dinagri 0/1	
Sec2		Sec2 is a dummy taking the value of 1 if the company	Binary 0/1	
	Eikon	belongs to the service sector (health services, consumer services, telecommunication, services to communities).		
		services, telecommunication, services to communities).		

Table 2: Descriptive statistics of the variables included in the regression analyses

The sample consists of all the SBF 120 Index firms over the period 2009-2012. This table reports the descriptive statistics of the variables included in the analyses. ESO is the percentage of shares held by employees. CEO age is the age of CEO in years. CEO tenure is the number of years in position. CEO turnover is a dummy variable taking the value of 1 if a CEO dismissal happens and 0 otherwise. ROE is the return on equity. Sales is the total sales of the company in euros. Own. Concentration measures the percentage of equity held by the largest shareholder. External director is the number of external directors divided by the total board members. Board size is the total number of board members. CEO duality is the dummy variable taking the value of 1 if the CEO is also the chairman and 0 otherwise. Dual structure takes the value of 1 if the company has a two tiered board structure and 0 otherwise. Family is a dummy variable taking the value of 1 if the company is owned by the family and 0 otherwise. State is the dummy variable taking the value of 1 if the state holds shares of the company and 0 otherwise. Tobins' Q is defined as the market value of equity at the end of the fiscal year plus the book value of assets minus the book value of equity, all divided by the book value of assets. Sec1 is a dummy taking the value of 1 if the company belongs to the industrial sector (raw material, industry, consumption goods). Sec2 is a dummy taking the value of 1 if the company belongs to the service sector (health services, consumer services, telecommunication, services to communities). For each variable, we compute the statistics when the level of ESO is below the median (<MED) and above the median (>MED). ***, **, * indicate coefficients significant at the 1%, 5%, and 10% levels, respectively.

coefficients sig	gnificant	at the 1%	, 5%, a	ınd 10% le	vels, resp	ectively.													
Panel A: Con	tinuous v	ariables																	
		N			MEAN		T-test	ME	DIAN (MI	ED)		SD			MIN.			MAX.	
	<med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td><td><med -<br="">>MED</med></td><td><med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td></med<></td></med<></td></med<></td></med<></td></med<></td></med<>	>MED	All	<med< td=""><td>>MED</td><td>All</td><td><med -<br="">>MED</med></td><td><med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td></med<></td></med<></td></med<></td></med<></td></med<>	>MED	All	<med -<br="">>MED</med>	<med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td></med<></td></med<></td></med<></td></med<>	>MED	All	<med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td></med<></td></med<></td></med<>	>MED	All	<med< td=""><td>>MED</td><td>All</td><td><med< td=""><td>>MED</td><td>All</td></med<></td></med<>	>MED	All	<med< td=""><td>>MED</td><td>All</td></med<>	>MED	All
CEO age	223	251	474	55.48	55.92	55.72	-0.7473	55	56	56	6.90	5.89	6.38	41	43	41	76	72	76
CEO tenure	223	251	474	4.84	4.58	4.71	0.72	4	4	4	4.01	3.74	3.87	0	0	0	17	17	17
ESO	223	222	445	.34	4.50	2.41	-11.5***	.23	2.59	1.03	.34	5.37	4.33	0	1.04	0	1.03	28.7	28.7
BEOR	214	211	476	0.7	3.7	2.2	-5.28***	0	0	0	3.86	8.15	6.5	0	0	0	28.54	36.84	36.84
Own. concentration	222	220	442	36.01	22.03	29.05	7.52***	31.44	17.4	26	20.89	18.05	20.72	4.42	2.5	2.5	78.96	84.51	84.51
External directors	223	221	444	37.64	45.07	41.34	-3.15***	32.34	42.55	37.81	23.09	26.41	25.05	0	0	0	100	94.76	100
Board size	223	221	444	11.73	13.54	12.63	-5.59***	11	14	12	3.23	3.55	3.51	4	3	3	21	23	23
ROE	207	239	446	11.55	9.59	10.50	2.09**	10.35	9.58	9.78	11.43	8.18	9.86	-17.04	-10.73	-17.04	66.38	38.1	66.38
Sales	223	252	475	6129969	2.2e+7	1.4e+7	-7.73***	1914300	9387000	4220410	1e+7	3e+7	2.4e+7	1119	121972	1119	6.7e+7	1.8e + 8	1.8e+8
Tobin's Q	198	227	425	1.16	.79	.97	4.75***	.85	.72	.77	1.07	.41	.81	.04	.054	0.04	8.02	2.40	8.02
Panel B: Dum	ımy varia	ables							Yes (x=	=1)						No (x=0))		
CEO turnover									9.09							90.91			
CEO duality									72.05	5						27.95			
Dual Structure	;								72.94	1						27.06			
Family									32.77	7						67.23			
State									13.24	1						86.76			
Sec1									63.87	7						36.13			
Sec2									53.78	3						46.22			

Table 3: Correlation matrix of the variables included in the regression analyses

The sample consists of all the SBF 120 Index firms over the period 2009-2012. This table reports the correlation matrix of the variables included in the analyses. ESO is the percentage of shares held by employees. CEO age is the age of CEO in years. CEO tenure is the number of years in position. CEO turnover is a dummy variable taking the value of 1 if a CEO dismissal happens and 0 otherwise. ROE is the return on equity. Sales is the total sales of the company in euros. Own. Concentration measures the percentage of equity held by the largest shareholder. External director is the number of external directors divided by the total board members. Board size is the total number of board members. CEO duality is the dummy variable taking the value of 1 if the CEO is also the chairman and 0 otherwise. Dual structure takes the value of 1 if the company has a two tiered board structure and 0 otherwise. Family is a dummy variable taking the value of 1 if the company is owned by the family and 0 otherwise. State is the dummy variable taking the value of 1 if the state holds shares of the company and 0 otherwise. Tobins' Q is defined as the market value of equity at the end of the fiscal year plus the book value of assets minus the book value of equity, all divided by the book value of assets. Sec1 is a dummy taking the value of 1 if the company belongs to the industrial sector (raw material, industry, consumption goods). Sec2 is a dummy taking the value of 1 if the company belongs to the service sector (health services, consumer services, telecommunication, services to communities). ***, **, * indicate coefficients significant at the 1%, 5%, and 10% levels, respectively.

	1	2	3	4	5	6	7	8	9	10	11	12	13	14
1. CEO age	1													
2. CEO tenure	0.05*	1												
3. ESO	0.05	-0.09	1											
4. ROE	-0.02	0.17***	-0.06	1										
5. Sales	0.22***	-0.03***	0.32***	-0.14*	1									
6. Own. concentration	0	-0.12**	-0.18***	0.05	-0.14**	1								
7. External directors	-0.04	0.16**	-0.06	-0.02	0.10***	-0.48***	1							
8. Board size	0.18***	-0.06**	0.18***	-0.05	0.49***	0.02	-0.28***	1						
9. BEOR	0.16*	-0.07**	0.03	-0.10**	0.24***	0.25**	-0.23***	0.37***	1					
10. Tobin's Q	0.01	0.1	-0.13**	0.27***	-0.26***	0.07***	-0.13***	-0.17***	-0.09	1				
11. CEO turnover	-0.07**	-0.34***	0.01	-0.15***	0.03	0.05	-0.07	-0.01	0.04	-0.05	1			
12. CEO duality	0.13	0.02	0.13	0.06	0.12*	-0.01	-0.17*	0.21***	0.12**	-0.005	-0.1*	1		
13. Dual Structure	-0.14**	-0.14***	-0.03	0.03	0.08**	-0.05	-0.08**	0.13***	-0.08	-0.07	-0.009	0.18***	1	
14. Family	-0.05	0.05	-0.20***	0.051	-0.29***	0.19***	-0.15***	-0.18***	-0.19***	0.2***	-0.06	-0.09***	0.09	1
15. State	0.09**	-0.03*	0.11*	-0.04	0.18	0.04	-0.09	0.35	0.52***	-0.1**	0.03	0.03	-0.005	-0.28***

Table 4: ESO and CEO entrenchment

This table reports the GMM estimates. The dependent variables are the CEO age, the CEO tenure and the CEO turnover. The sample consists of all the SBF 120 Index firms over the period 2009-2012. ESO is the percentage of shares held by employees. ROE is the logarithm of the return on equity. Sales is the logarithm of the total sales of the company in euros. Own. Concentration measures the percentage of equity held by the largest shareholder. External director is the number of external directors divided by the total board members. Board size is the total number of board members. CEO duality is the dummy variable taking the value of 1 if the CEO is also the chairman and 0 otherwise. Dual structure takes the value of 1 if the company has a two tiered board structure and 0 otherwise. Family is a dummy variable taking the value of 1 if the company is owned by the family and 0 otherwise. State is the dummy variable taking the value of 1 if the state holds shares of the company and 0 otherwise. Tobins' Q is defined as the market value of equity at the end of the fiscal year plus the book value of assets minus the book value of equity, all divided by the book value of assets. Sec1 is a dummy taking the value of 1 if the company belongs to the industrial sector (raw material, industry, consumption goods). Sec2 is a dummy taking the value of 1 if the company belongs to the service sector (health services, consumer services, telecommunication, services to communities). The table presents the coefficients, robust standard errors t-values and then the adjusted R². We report R² of the fixed-effects OLS regressions. N is the number of non-missing observations in the sample. ***, **, * indicate coefficients significant at the 1%. 5%, and 10% levels, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	CEO age	CEO age	CEO age	CEO	CEO	CEO	CEO	CEO	CEO
	CLO uge			tenure	tenure	tenure	turnover	turnover	turnover
ESO		0.09	-1.57*		-0.02	0.16***		-0.02**	0.006**
		(0.30)	(-1.87)		(0.02)	(0.03)		(0.01)	(0.003)
ESO ²			0.081**			-0.007***			-0.0004*
			(2.18)			(0.002)			(0.0002)
Own.	0.12*	0.12*	0.07	-0.01	-0.01***	0.01	-0.001	-0.003**	-0.001***
concentration	(0.07)	(0.07)	(1.27)	(0.01)	(0.003)	(0.02)	(0.001)	(0.001)	(0.0002)
External		, ,		, ,			, , ,	, ,	
directors	0.28***	0.28***	0.20***	0.02**	0.01***	0.03**	-0.002	-0.004	-0.0004
	(0.10)	(0.09)	(2.71)	(0.01)	(0.005)	(0.01)	(0.003)	(0.003)	(0.001)
Board size	6.80*	6.28***	3.91***	-0.004	0.02	0.04***	-0.013	-0.01	-0.02
	(1.98)	(1.71)	(3.64)	(0.08)	(0.02)	(0.01)	(0.01)	(0.01)	(0.02)
CEO duality	-6.40***	-5.48*	-3.27	1.00*	1.10***	1.03***	-0.02***	-0.03***	0.16**
	(3.62)	(3.14)	(-1.27)	(0.54)	(0.17)	(0.10)	(0.006)	(0.001)	(0.08)
Dual Structure	-4.20	-4.23*	-3.22	-1.52***	4.40***	-1.31***	0.06***	0.01	0.01
	(2.77)	(2.24)	(-1.43)	(0.52)	(0.22)	(0.05)	(0.01)	(0.02)	(0.03)
Family	-1.96	-1.76	-1.43	0.0006	-0.13	-0.39***	-0.08***	-0.08***	-0.04***
•	(3.02)	(2.58)	(-0.93)	(0.56)	(0.25)	(0.14)	(0.01)	(0.006)	(0.003)
State	-16.62***	-14.30***	-8.29**	0.31	0.19	-0.04	0.06	0.08	0.22
	(6.47)	(2.74)	(-2.00)	(0.75)	(0.42)	(0.35)	(0.11)	(0.12)	(0.20)
ROE	-2.03	-2.23	-1.57	0.56*	0.48***	0.50***	-0.09***	-0.07**	-0.06***
	(1.62)	(1.69)	(-1.02)	(0.34)	(0.18)	(0.13)	(0.03)	(0.04)	(0.007)
Sales	-6.32***	-5.89***	-2.76**	-0.13	-0.17***	-0.23***	-0.0008	0.02	-0.003
	(2.26)	(2.02)	(-2.11)	(0.19)	(0.04)	(0.06)	(0.01)	(0.02)	(0.01)
Tobin's Q	2.17	2.06	1.54***	0.39	0.47***	0.48***	0.04***	0.01**	0.03***
	(1.76)	(1.62)	(0.98)	(0.32)	(0.08)	(0.04)	(0.001)	(0.007)	(0.001)
Sec1	12.90***	13.19***	10.18	1.52**	1.65***	2.03***	-0.09	-0.10	-0.10
	(4.60)	(4.22)	(3.01)	(0.69)	(0.29)	(0.48)	(0.11)	(0.10)	(0.07)
Sec2	14.33***	13.36***	10.22***	-0.45	-0.15	0.10	-0.09	-0.04	-0.12
	(5.32)	(4.72)	(2.64)	(0.67)	(0.10)	(0.27)	(0.13)	(0.10)	(0.11)
Constant	51.36***	50.80***	39.39***	4.11	-1.52***	3.34**	0.66**	0.46*	0.50***
	(15.96)	(14.83)	(2.85)	(3.00)	(1.26)	(1.56)	(0.33)	(0.24)	(0.054)
R ²	0.16	0.16	0.16	0.15	0.15	0.15	0.06	0.06	0.06
J-statistique									
(p-value)	0.32	0.29	0.05	0.59	0.20	0.33	0.36	0.45	0.63
Nb of	10	20	22	10	20	10	1.0	10	22
instruments	18	20	23	18	20	19	18	18	23
Observations	229	238	149	238	229	230	149	149	148

Table 5: ESO, CEO entrenchment and representation of employee owners on the board

This table reports the GMM estimates. The dependent variables are the CEO age, the CEO tenure and the CEO turnover. The sample consists of all the SBF 120 Index firms over the period 2009-2012. ESO is the percentage of shares held by employees. BEOR is the proportion of employee owners seating on the board. ROE is the logarithm of the return on equity. Sales is the logarithm of the total sales of the company in euros. Own. Concentration measures the percentage of equity held by the largest shareholder. External director is the number of external directors divided by the total board members. Board size is the total number of board members. CEO duality is the dummy variable taking the value of 1 if the CEO is also the chairman and 0 otherwise. Dual structure takes the value of 1 if the company has a two tiered board structure and 0 otherwise. Family is a dummy variable taking the value of 1 if the company is owned by the family and 0 otherwise. State is the dummy variable taking the value of 1 if the state holds shares of the company and 0 otherwise. Tobins' Q is defined as the market value of equity at the end of the fiscal year plus the book value of assets minus the book value of equity, all divided by the book value of assets. Sec1 is a dummy taking the value of 1 if the company belongs to the industrial sector (raw material, industry, consumption goods). Sec2 is a dummy taking the value of 1 if the company belongs to the service sector (health services, consumer services, telecommunication, services to communities). The table presents the coefficients, robust standard errors t-values. We report R² of the fixed-effects OLS regressions. N is the number of non-missing observations in the sample. ***, **, * indicate coefficients

significant at the 1%, 5%, and 10% levels, respectively.

	(10)	(11)	(12)		
	CEO age	CEO tenure	CEO turnover		
ESO	-2.36***	-0.32***	-0.03***		
	(0.17)	(0.04)	(0.01)		
ESO ²	0.07***	0.02**	0.001***		
	(0.00)	(0.00)	(0.00)		
BEOR	-62.19***	-11.19***	-1.82***		
	(25.45)	(1.32)	(0.51)		
BEOR*ESO	8.89	6.37***	0.96***		
	(13.13)	(0.39)	(0.22)		
BEOR*ESO2	-0.23	-0.25***	-0.04***		
	(0.52)	(0.00)	(0.01)		
Own.		` '	• • •		
concentration	-0.16***	0.01	-0.003***		
concentration	(0.00)	(0.02)	(0.001)		
External	, ,		· · · · ·		
directors	-0.02	0.08***	-0.005		
unectors	(0.12)	(0.02)	(0.003)		
Doord size	1.24	0.11**	-0.03		
Board size					
CEO 1 -1'	(1.19)	(0.05)	(0.02)		
CEO duality	11.50	1.41***	-0.01*		
7 . 10.	(16.04)	(0.16)	(0.01)		
Dual Structure	-39.22***	-1.44***	-0.05***		
	(7.85)	(0.35)	(0.01)		
Family	4.87***	0.32***	-0.06***		
	(1.31)	(0.12)	(0.03)		
State	-1.46	0.70*	0.10		
	(5.07)	(0.38)	(0.10)		
ROE	1.27***	0.27***	-0.09***		
	(0.07)	(0.07)	(0.03)		
Sales	1.24	-0.33***	0.02		
	(1.32)	(0.03)	(0.02)		
Tobin's Q	-3.46***	0.44***	-0.01***		
-	(0.26)	(0.14)	(0.00)		
Sec1	2.33**	2.04***	-0.13		
	(1.07)	(0.69)	(0.11)		
Sec2	6.19***	0.02	-0.11		
	(0.02)	(0.29)	(0.12)		
Constant	31.23***	2.47	0.71***		
Constant	(7.78)	(3.20)	(0.26)		
	(1.10)	(3.20)	(0.20)		
R ²	0.16	0.15	0.07		
J-statistique (p-					
value)	0.06	0.18	0.67		
Nb of					
instruments	19	28	24		
Observations	150	148	150		